Author's response to reviews

Title: Species diversity of non-tuberculous mycobacteria isolated from humans, livestock and wildlife in the Serengeti ecosystem, Tanzania

Authors:

- Bugwesa z Katale Dr (bugwesa2002@yahoo.co.uk)
- Erasto V Mbugi Dr (rerasto@yahoo.com)
- Louise Botha Mrs (louisev@sun.ac.za)
- Julius D Keyyu Dr (julius.keyyu@tawiri.or.tz)
- Sharon Kendall Dr (skendall@RVC.AC.UK)
- Hazel Docrell Prof (Hazel.Docrell@lshtm.ac.uk)
- Anita L Michel Prof (Anita.Michel@up.ac.za)
- Rudovick R Kazwala Prof (kazwala@gmail.com)
- Mark M Rweyemamu (mark.rweyemamu@btinternet.com)
- Paul van Helden (PVH@sun.ac.za)
- Mecky I Matee (mateemecky@yahoo.com)

Version: 3 Date: 16 July 2014

Author's response to reviews:

Bugwesa Zablon Katale
Muhimbili University of Health and Allied Sciences
P.O. BOX 2965
Dar es Salaam
16th July, 2014

The Editor,
BMC Infectious Diseases,
BioMed Central,
236 Gray's Inn Road,
London WC1X 8HB,
United Kingdom.
Dear BMC Editor,

RE: RESUBMISSION OF A MANUSCRIPT TITLED SPECIES DIVERSITY OF NON-TUBERCULOUS MYCOBACTERIA ISOLATED FROM HUMAN, LIVESTOCK AND WILDLIFE IN THE SERENGETI ECOSYSTEM, TANZANIA

Refer to the heading above.

I hereby resubmit a manuscript titled Species diversity of non-tuberculous mycobacteria isolated from humans, livestock and wildlife in the Serengeti ecosystem, Tanzania. This is to respond to comments from the editor to include
line numbering in the main text file of the manuscript.
Response: Line numbering and page numbers have been included in the main
text of the manuscript as advised by the editor. i.e. Lines 1 – 510 from page 1 to
22.
It is my sincere hope that the manuscript will be considered for publication.
Thank you for your cooperation.

Yours faithfully,

Bugwesa